Author's response to reviews

Title: New non-cognitive procedures for medical applicant selection: a qualitative analysis in one school.

Authors:

   Sara S Katz (sara10@012.net.il)
   Shlomo Sh Vinker (shlomovis@clalit.org.il)

Version: 5
Date: 5 October 2014

Author's response to reviews: see over
Dear Mr Aldrin Ulep,

On behalf of Prof Sandra Carr

Here are our changes according to your numbered comments, from Oct 03 MS:
1159315833133566 – New non-cognitive procedures for medical applicant selection: a qualitative analysis in one school.

1. I moved the ethics section and placed it in the methods section. Lines: 269-272

2. The sentence was removed from the ethics section. See lines 269-272.

3. Thanks for changing the article type into: "a research article"

4. I included a statement on RATS guidelines, lines: 248-249.

5. I noticed a mistake in our names. We don't have a second name.

Thank you very much
Sara Katz and Shlomo Vinker.